Evidence of Local Persistence of Human Anthrax in the Country of Georgia Associated with Environmental and Anthropogenic Factors

Ian T. Kracalik^{1,2}, Lile Malania³, Nikoloz Tsertsvadze³, Julietta Manvelyan³, Lela Bakanidze⁴, Paata Imnadze³, Shota Tsanava³, Jason K. Blackburn^{1,2}*

1 Spatial Epidemiology and Ecology Research Lab, Department of Geography, University of Florida, Gainesville, Florida, United States of America, 2 Emerging Pathogens Institute, University of Florida, Gainesville, Florida, United States of America, 3 National Center for Disease Control and Public Health, Tbilisi, Georgia, 4 Agrarian University of Georgia, Tbilisi, Georgia

Abstract

Background: Anthrax is a soil-borne disease caused by the bacterium *Bacillus anthracis* and is considered a neglected zoonosis. In the country of Georgia, recent reports have indicated an increase in the incidence of human anthrax. Identifying sub-national areas of increased risk may help direct appropriate public health control measures. The purpose of this study was to evaluate the spatial distribution of human anthrax and identify environmental/anthropogenic factors associated with persistent clusters.

Methods/Findings: A database of human cutaneous anthrax in Georgia during the period 2000–2009 was constructed using a geographic information system (GIS) with case data recorded to the community location. The spatial scan statistic was used to identify persistence of human cutaneous anthrax. Risk factors related to clusters of persistence were modeled using a multivariate logistic regression. Areas of persistence were identified in the southeastern part of the country. Results indicated that the persistence of human cutaneous anthrax showed a strong positive association with soil pH and urban areas.

Conclusions/Significance: Anthrax represents a persistent threat to public and veterinary health in Georgia. The findings here showed that the local level heterogeneity in the persistence of human cutaneous anthrax necessitates directed interventions to mitigate the disease. High risk areas identified in this study can be targeted for public health control measures such as farmer education and livestock vaccination campaigns.

Citation: Kracalik IT, Malania L, Tsertsvadze N, Manvelyan J, Bakanidze L, et al. (2013) Evidence of Local Persistence of Human Anthrax in the Country of Georgia Associated with Environmental and Anthropogenic Factors. PLoS Negl Trop Dis 7(9): e2388. doi:10.1371/journal.pntd.0002388

Editor: Stephen Baker, Oxford University Clinical Research Unit, Viet Nam

Received January 11, 2013; Accepted July 17, 2013; Published September 5, 2013

Copyright: © 2013 Kracalik et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This project was funded by the U.S. Defense Threat Reduction Agency's Cooperative Biological Engagement Program. The funder had no role in the study design, data collection or analysis. A final draft of this manuscript was approved by a formal inter-agency review organized by the funder as a requirement of the funding.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: jkblackburn@ufl.edu

Introduction

Recently it has been suggested that the burden of anthrax has not been fully realized [1]. Interest in anthrax as a biological weapon has grown since the 2001 bioterrorist attacks [2] yet the disease is often undervalued as a public health threat. Anthrax is considered a neglected zoonosis, disproportionately afflicting rural areas in developing nations [3]. Although several countries have implemented successful control strategies, the disease continues to persist globally with an estimated 20,000 to 100,000 new cases yearly [4]. However, the true disease burden is likely unknown. Areas most heavily impacted by the disease include countries of sub-Saharan Africa, and former Soviet states in Southwestern and Central Asia [5,6].

The causative agent, *Bacillus anthracis*, is a soil-borne, Grampositive bacterium, which primarily infects herbivores and secondarily afflicts humans. The bacterium is able to persist in the environment for years, possibly even decades, in alkaline soils with high calcium and organic matter [7] and has been shown to be limited geographically by other environmental factors such as temperature [8,9,10]. Human transmission is often a result of coming into contact with infected animals or contaminated animal materials during agricultural activities including the butchering of livestock or industrial exposures through the processing of hair and bone [8,9,10]. Manifestation of the disease in humans typically occurs in three acute clinical forms with cutaneous anthrax comprising $\sim 95\%$ of all reported cases, followed by gastrointestinal and inhalation anthrax [8,9,10]. Although the most common form of the disease is treatable with antimicrobials, if left untreated, human cutaneous anthrax can be highly fatal with case fatality ratios in excess of 20% [11,12]. In neglected endemic regions anthrax can result in substantial economic losses from livestock mortality and lost worker productivity.

Author Summary

Anthrax is a zoonotic bacterial disease that occurs nearly worldwide. Despite a large number of countries reporting endemic anthrax, persistence of the disease appears to be associated with specific ecological factors related to soil composition and climatic conditions. Human cases are most often associated with handling infected livestock or contaminated meat and most cases are in cutaneous form (skin infections). Following the collapse of the Soviet Union, the country of Georgia has undergone major restructuring in land management and livestock handling and anthrax remains a serious public health risk. Few studies have evaluated the local spatial patterns of human anthrax. Here we identify areas on the landscape where human cutaneous anthrax persisted over the last decade. Persistence was found to be associated with both anthropogenic and environmental factors including soil pH and livestock density. These findings aid in the establishment of spatial baseline estimates of the disease and allow public health officials to adopt targeted anthrax control strategies, such as livestock vaccination campaigns and farmer education.

In the country of Georgia, anthrax is classified as endemic [13] and has persisted for centuries with the first description of human disease thought to be anthrax documented in 1697 [4]. Following the collapse of the Soviet Union, public and veterinary health services in Georgia experienced tremendous setbacks due to deteriorating organization and finances [14]. During this period of transition to independence Georgia experienced an increase in the number of reported human anthrax cases with 118 reported cases during 1991 to 1996 compared to 36 cases reported during 1985 to 1990 [15]. Recent evidence indicates that the situation has continued to worsen and that incidence rates have surpassed that of neighboring countries, including hyperendemic Turkey [5]. Although human anthrax outbreaks are in some instances sporadic, as evident by the recent outbreak in Bangladesh after a more than 20 year absence [14,16], the ability of the bacterium to survive in the environment [16,17] can give rise to areas of persistence and disease recurrence. Additionally, the heterogeneous nature of the disease [18] necessitates control efforts that target high risk areas rather than employing blanket control efforts. This is especially true in developing nations where resources are often limited. Therefore, identifying local areas of persistent may allow for the more efficient implementation of public health intervention strategies such as livestock vaccination, increased awareness of the disease, and placement of improved diagnostics for local veterinary and human health care facilities.

This study had three objectives: 1) to analyze human anthrax at a local scale to better understand the distribution and burden of disease over the past decade, 2) identify areas of persistence for targeted public health control measures 3) identify environmental and anthropogenic factors associated with areas of persistence.

Methods

Ethics Statement

No human subjects work was undertaken in this study, human anthrax case data were extracted from government reports prepared and approved by the Georgian National Center for Disease Control and Public Health (NCDC). These government reports provided summarized count data of patients diagnosed at health care facilities by category of disease and year. All data were anonymised.

Data Collection and Management

Anthrax is a nationally reportable infectious disease in Georgia. Surveillance and documentation within the country is undertaken by the NCDC, which is comprised of a reporting network of 11 regional and 66 district public health centers [7,19]. A spatial database of reported human cutaneous anthrax was constructed with a geographic information system (GIS) using case data from 2000 to 2009 (Figure S1). Latitude and longitude coordinate pairs were matched to the reported community using the National Geospatial Intelligence Agency GEOnet names server [20], Index Mundi [21], and the GeoNames database [22]. Population data for each community were derived from the Population Statistics of Eastern Europe database [23] for the year 2002. Of the communities that were included in this human anthrax database we were unable to identify population estimates for seven of these locations. In these instances, the population of these seven communities were estimated using the Gridded Population of the World (GPW3) population dataset for the year 2000 [24]. Yearly population estimates were extrapolated using the United Nations (UN) medium variant population growth rates for Georgia during the periods 2000 to 2005 (-1.17) and 2005 to 2010 (-0.57) (http:// data.un.org/) using the following formula:

 $e^{ab}(p)$

where *a* is the UN population growth rate for either the period 2000 to 2005 or 2005 to 2010 aforementioned, *b* is the number of years from the reference population and (p) is the population.

Epidemiological Analysis

Yearly national incidence rates per one million population were calculated with the total number of cases reported in Georgia for each year as the numerator and the yearly national population obtained from GeoStat [25] as the denominator. The study period was divided into two equal five-year periods to identify potential changes in the reporting of the disease over time and space. Cumulative incidence risk was calculated per community for each five-year period (2000 to 2004) and (2005 to 2009) with the total number of cases during each period as the numerator and the median year community population of each period as the denominator. Cumulative incidences per 10,000 population were mapped at the community level and at the district level using ArcGIS 9.3.1 [26]. The methodology for mapping at the district level is described in Text S1. The average incidence per 10,000 population was calculated to investigate the global clustering among communities. Smoothed risk estimates were calculated for each five-year time period using the Empirical Bayes Smoothing (EBS) in the GeoDa software package [27]. The EBS technique can be used to adjust for instability in the risk estimates caused by heterogeneity in the distribution of cases and the population. Posterior risk is estimated from a weighted combination of the local risk and the risk in the surrounding areas (the prior). It has been suggested that the EBS methodology can be implemented in several scenarios, such as when the numerator data total less than three cases, which was the situation in this analysis [28]. In order to maintain a standard comparison of rates between time periods, crude and EBS estimates of cumulative incidence rates (per 10,000) were mapped using graduated symbols with the same data bins (0-3, 4-9, 10-18, 19-30, 31-50, >50). A chi-square analysis

was used to test for a relationship between the number of communities reporting anthrax in the east and west of the country during each time period. Boxplots were used to illustrate differences between the crude and EBS rates for both of the five-year time periods.

Spatial Autocorrelation Analysis

To test for spatial dependence in EBS average anthrax incidence rates between communities, the global Moran's I test was implemented using OpenGeoDa [29]. This test analyzes the degree of spatial autocorrelation present in the data in relation to a single variable of interest and is written following Moran [30]:

$$I = \frac{N}{\sum_{i} \sum_{j} w_{ij}} \frac{\sum_{i} \sum_{j} (Xi - \bar{X})}{\sum_{i} (Xi - \bar{X})^{2}}$$

where N is the number of communities, \bar{X} is the average EBS incidence at each community, X_i and X_j are the EBS incidence in community *i* and *j*, and W_{ij} is the spatial weight matrix defining the spatial relationships between communities. In this instance W_{ij} was defined as a series of increasing distances from 1 to 10 kilometers (km) to test a range of dependencies. The statistic produces values between -1 and +1 similar in interpretation to the Pearson's correlation with large positive values of *I* indicating the similarity of values between locations with the presence of either positive or negative autocorrelation and negative values indicating a dissimilarity among values. Significance of the test statistic was assessed with a pseudo p-value generated using 999 random permutations [29].

Spatial and Space-Time Cluster Analysis

To identify the spatial and spatio-temporal clustering of anthrax cases during the 10-year period the Poisson clustering models using the space-only and space-time scan statistics were implemented in SaTScan v9.0 [31]. SaTScan uses a series of moving windows of varying diameter to detect spatial clusters across a study area and simultaneously implements a series of cylinders of varying height to detect temporal clustering as well [29]. The relative risk (RR) is calculated based on the number of observed and expected observations and the likelihood function for each window location and size depending on the assumed distribution [32]. For the SaTScan analyses, community latitude and longitude coordinates were used as the case locations. The model was run on yearly case data, using the total number of yearly cases per community while adjusting for the underlying population of each community. Space-only models were run for each five-year time period using 25% of the population at risk. This threshold for the population at risk was chosen to identify local clusters of anthrax rather than using the default of 50%. Maximum spatial extents of 25%, 20%, 15%, and 10% of the population at risk were chosen with no change in cluster size or location, therefore 25% was used in the analysis. All SaTScan cluster were chosen at the $p \le 0.05$ level of significance.

In this study we were particularly interested in identifying areas of human anthrax that experienced recurrent infections or persistence over space and time. To identify persistence we implemented the Space-Time Poisson model in SaTScan using 25% of population at risk with a maximum temporal window of 90% of the study period. The temporal window was extended to 90% to allow SaTScan to search across a greater portion of the study period and thereby identify communities that may have persisted as clusters in both space and time across a large temporal window. Additionally, a SaTScan model examining spatial variation in a temporal trend was used to identify individual communities that may have been contributing to an increasing trend in incidence over time. SaTScan fixes the temporal window during the analysis and compares the trend in risk inside and outside of the varying size spatial window [32,33]. The most likely spatial cluster location is chosen by the cylinder that maximizes the change in the trend over time.

Environmental and Anthropogenic Variables

Environmental and anthropogenic variables used in this analysis have been suggested in the literature to be associated with the presence of human anthrax (Table 1) [32,33]. Continuous environmental variables included elevation, average annual precipitation [32], mean land surface temperature (LST), mean mid infra-red reflectance (MIR) [3,6,7,8], and minimum soil pH (soil pH) [34]. Continuous anthropogenic variables included population density [35], travel time to cities with a population >50,000 (travel time) [36] and categorical variables for urban/ rural classification (UR) [26] and cattle density dichotomized into equal groups <19 head per 1 km² and >19 head per 1 km² (cattle density) [37].

Logistic Regression

Logistic regression was used to identify differences in risk factors between communities that were identified as a persistent spacetime cluster in SaTScan and those communities that were not. A total of nine environmental/anthropogenic variables (Table 1) were used to construct univariate, and a final multivariate, logistic regression models using a binary response 0/1 with clustered communities = 1 (n = 24) and non-clustered communities = 0 (n = 80) as the dependent variable. The final model was constructed in SAS using a backward stepwise approach with entry p-values of 0.15 and stay p-values of 0.20. Variables used in the stepwise model selection were limited to those that did not show significant collinearity ($\rho_s \leq 0.70$) using a Spearman's rank correlation test. Goodness of fit was evaluated with the Hosmer and Lemeshow test and interaction terms were tested between categorical and continuous independent variables. To assess the discrimination ability of the final logistic regression model we used receiver operating (ROC) curves, based on the area under the curve (AUC) [38]. The AUC provides a measure of model accuracy with values ranging from 0.0 to 1, where values of 1 indicates all locations were correctly classified by the model and 0.5 indicating no better than random.

Results

National Incidence

During the study period there were 340 cases of human anthrax reported in Georgia with a median number of reported cases per year of approximately 33.5 (CI_{95%}: 22.5, 42.0). Yearly national incidences per one million displayed inter annual variability ranging from a low of 3.4 (CI_{95%}: 1.9, 5.7) in 2002 and to a high of 13.9 (CI_{95%}: 10.7, 17.9) in 2008 (Figure 1). There were 143 cases [range: 15–45] during period one (2000 to 2004) with a median number of yearly cases of 27 (CI_{95%}: 9, 39) and a total of 197 cases [range: 18–61] during period two (2005 to 2009) for a yearly median of 38 (CI_{95%}: 15, 58).

Community Level Analysis

Figure 2 shows the spatial distribution of crude (insets 1A and 2A) and smoothed (insets 1B and 2B) community level cumulative

Table 1. Environmental and anthropogenic variables used in the logistic regression model.

Environmental Variables	Rationale
Mean Annual Precipitation	Precipitation has been shown to be associated with the distribution of the bacterium and seasonal outbreak events.
Elevation	Low lying elevations may act as storage areas for the pathogen as a result of surface runoff from storms.
Mean Mid Infra-Red Reflectance	Soil moisture is suggested to be associated with the presence of the bacterium.
Mean Land Surface Temperature	The bacterium prefers a temperature range between 18° and 39°C.
Minimum Soil pH	Alkaline soils with a $pH>6.0$ are considered favorable for the pathogen.
Anthropogenic Variables	
Cattle Density	Areas with higher cattle density would likely have more agriculture and a greater chance of coming into contact with infected animals.
Urban/Rural	Rural areas often experience a higher burden of disease due to the frequency of peridomestic living arrangements.
Travel Time to City $>$ 50,000	Longer travel times are more indicative of rural areas and higher agricultural intensity.
Population Density	Lower population density may be associated with rural areas and agricultural production.

doi:10.1371/journal.pntd.0002388.t001

incidence rates in Georgia for each time period, respectively. There were a greater number of communities reporting human anthrax during period two (n = 79) compared to period one (n = 44) with an increased number occurring in the west of the country during period two (n = 47) compared to period one (n = 23); however this difference was not significant ($\chi^2 = 0.6$, p = 0.44). Community level cumulative incidence rates per 10,000 population during period one ranged from a low of 0.09 (CI_{95%}: 0.05,

0.17) to a high of 136.4 ($CI_{95\%}$: 83.3, 210.07) and during period two ranged from a low of 0.18 ($CI_{95\%}$: 0.11, 0.29) to a high of 242.3 ($CI_{95\%}$: 97.07, 499.2). Box plots illustrated that the cumulative incidence rates for each time period were unstable due to population heterogeneity with a greater than 30% reduction in the estimates applying an EBS rate adjustment (Figure S2). District level estimates of cumulative risk are displayed in Figure S3.



Figure 1. National reporting of human cutaneous anthrax in Georgia by year and incidence per 1 million population. doi:10.1371/journal.pntd.0002388.g001



Figure 2. Maps showing the cumulative incidences of human cutaneous anthrax in Georgia at the community level. Symbol sizes represent cumulative incidence risk per 10,000 population for period 1 (1A) and period 2 (2A). Empirical Bayes Smoothing estimates were calculated for both period 1 (1B) and period 2 (2B) to adjust for population heterogeneity. doi:10.1371/journal.pntd.0002388.g002

Cluster Analysis

The Moran's I test identified the presence of significant positive spatial autocorrelation (Figure 3). A correlogram of Moran's Iindex values showed increases in positive values up to a distance of 3000 meters and then a subsequent sharp decline up to a distance of 8000 meters indicating spatial dependence at a relatively local scale. The space-only SaTScan analysis detected the presence of three significant clusters during period one (Figure 4A) and two clusters during the period two (Figure 4B). Spatial clusters in period one were smaller, and contained fewer communities compared to period two. Clusters varied spatially between periods with a large cluster in period two located in the center of the country that was absent during period one. Persistent clusters in both periods were identified around the main urban center, the capital Tbilisi in the southeast.

Results from the space-time model identifying persistence revealed the presence of four significant persistent clusters with two cluster located near the capital Tbilisi (Figure 5A). Clustered communities had an average of 1.01 cases per year compared to 0.2 cases per community outside of clusters. Clusters were predominantly located in the central and eastern part of the country excluding a large number of communities in the west. The spatial variation in a temporal trend model indicated a single cluster of communities that were contributing to an increasing trend in human anthrax reporting (Figure 5B). Communities within the central part of the country were shown to have incidence of anthrax increasing at an average of 38.1% per year as compared to 2.8% per year for the rest of the communities in the study area.

Logistic Regression Analysis

Results from the logistic regression models are shown in Table 2. The Hosmer and Lemeshow test (p = 0.96) revealed the model was a good fit to the data. There were no statistically significant interaction terms at the $p \le 0.10$. Accuracy of the final main effects model evaluated by the AUC from the ROC analysis showed good discrimination (AUC = 0.85, p < 0.001). Environmental and anthropogenic factors were both found to be associated with the presence of persistence. Strong positive associations shown in odds ratios (OR) were found between soil pH (OR 4.58, CI_{95%}: 1.55, 13.51), and UR (urban areas have greater odds OR, 4.67 CI_{95%}: 1.11, 19.64). Cattle density (cattle density above 19 head per 1 km² have greater odds, OR 2.78, CI_{95%}: 0.91, 9.55) was also shown to be positively associated although not at the 0.05 level of significance (p = 0.10).

Discussion

Anthrax in the country of Georgia continues to represent a threat to public and veterinary health. The current national



Figure 3. Spatial correlogram displaying global Moran's / values at varying distance thresholds. Distances ranged from 1,000 m to 10,000 m by 1,000 m increments. Dashed lines represent 95% confidence intervals for the Moran's / values. doi:10.1371/journal.pntd.0002388.g003

classification of endemicity does not allow for an efficient implementation of control measures given the limited resources available locally. Previous human studies have focused on national reporting [39], investigations of a single outbreak, or outbreaks across a short period of time [40]. In contrast, this study represents a fine scale spatial investigation of naturally acquired human anthrax during a decade of reporting. We identified persistent clusters of disease at the community level that were associated with both environmental and anthropogenic factors. Establishing these sub-national estimates of human anthrax risk provides a crucial first step in implementing targeted public health interventions.

The high rates of anthrax observed during the study period are likely due to a combination of factors resulting from decreased funding for public and veterinary health management following the collapse of the Soviet Union as well as changes to agricultural production [17]. Since the transition to independence in the 1990s, reliance on agriculture has grown with an increase in agricultural employment from 25% in 1990 to ~55% in 2009 [18,41,42]. During this period agricultural decollectivization has marked a switch to individual ownership [5,43]. This has resulted in a greater number of individual animal holdings and a subsequent increase in peridomestic arrangements [43]. Greater contact among livestock and humans living in close proximity (peridomesticity) has been shown elsewhere to facilitate disease transmission [44]. The increases in reporting observed here may have been due to increased diagnostic capacity and awareness, although the issue of anthrax as a neglected, endemic disease [43], along with funding cuts to public health facilities, has likely led to an under reporting in both humans and animals [45]. In Georgia the number of documented livestock cases are few, with the number of human cases out of proportion suggesting an anthropocentric reporting system [3]. Additionally, the absence of gastrointestinal cases of anthrax during the study period, given the large number of cutaneous cases, further highlights the probable under reporting of the disease [5].

Results from this study point to a primarily localized transmission of human anthrax identified by the strong spatial dependence of the Moran's I statistic at a relatively short distance (3000 meters) (Figure 3). This is consistent with research that has described the local dynamics of infection related to the communal handling, butchering, or sharing of infected meat across short distances [14], although imported cases from contaminated animal products have been documented elsewhere [46]. In Georgia, anthrax was spatially heterogeneous at the community level as evident by the presence of clustering (Figure 4). These findings are similar to studies in livestock/wildlife, which are often the source of human infection, that found variation in cases related to environmental constraints on the pathogen [42,47,48]. Reports on the spatial distribution of human cases are largely absent in the literature although in Kazakhstan human reporting was shown to be located in specific areas that comprised historical foci of the disease [49]. By expanding the maximum temporal window to 90% in the



Figure 4. Spatial clustering of human cutaneous anthrax. Village level clustering was identified during the period 2000 to 2004 (A) and 2005 to 2009 (B). Star symbols represent cluster centers, red dots represent villages that were part of spatial cluster, and grey dots represent villages that were not part of a spatial cluster. Buffers of clusters are represented in kilometers (km). doi:10.1371/journal.pntd.0002388.g004

SaTScan space-time analysis we identified areas that represented anthrax persistence over time. The possibility persistence is likely due to a combination of ecological factors that permit pathogen survival and anthropogenic factors that accommodate disease transmission [20,48,50].

Soil pH showed a strong positive association with anthrax persistence, as documented elsewhere [51]. Additional environmental factors MIR and LST were also significant in the model. Alkaline soils with adequate moisture (MIR) and an ambient temperature above 15° C (LST) have been suggested as factors related to incubator areas for bacterium [8,52,53]. These factors are similar to previous studies that have modeled the potential distribution *B. anthracis* in the United States and Kazakhstan [7,8,50]. Soil pH has long been considered an important factor for

spore survival with the bacterium preferring alkaline soils with a relatively high pH [7]. In Tanzania, a higher proportion of anthrax seropositive wildlife and domestic animals were identifed in areas with alkaline soils [8,10]. Arable land is limited in Georgia (\sim 17% of all land), leaving pasture/grasslands for livestock grazing in more alkaline soils [7,54,55]. Cattle density had a positive association OR 2.78 (CI_{95%}: 0.91, 9.55) with persistence, however, it was not significant. Infection ratios comparing humans to livestock can vary widely, ranging from 1:3 to 1:16 as seen in Zambia [50] and the density of cattle may not be as important as the presence or absence of livestock.

Human anthrax has been described as a disease of rural communities [43]. However, our study suggests that urban areas had >4.5 times the odds of being a persistent cluster when



Figure 5. Distribution of space-time clusters defined as persistence and the spatial variation in a temporal trend model. Star symbols in (map inset A) represent the center of clusters defined as persistence, colored dots correspond to villages that were included within each specific space-time clusters, grey dots represent villages not part of a cluster, relative risk for each cluster is displayed (RR) along with the buffer size in kilometers (Km). Red dots in inset B refer to villages within the buffer that are increasing at a faster rate (38.1% annually) compared to those outside the buffer shown in grey dot (2.8% annually). doi:10.1371/journal.pntd.0002388.g005

 Table 2. Results of logistic regression examining environmental and anthropogenic factors associated with human anthrax clusters.

Variables*	Full Model (OR)	95% CI	Final Model (OR)	95% CI	χ^2 (p-value)
Mean Annual Precipitation	1.00	0.99, 1.01	-	-	
Elevation	1.00	0.99, 1.00	-	-	
Mean Mid Infra-Red Reflectance	1.00	0.96, 1.05	1.04	1.01, 1.08	0.03
Mean Land Surface Temperature	0.89	0.50, 1.58	0.65	0.40, 0.82	<0.01
Minimum Soil pH	3.31	0.89, 12.34	4.58	1.55, 13.51	<0.01
Cattle Density <19 (Ref)	1.00	-	1.00	-	0.10
>19	2.41	0.62, 9.44	2.78	0.91, 9.55	
Rural (Ref)	1.00	-	1.00	-	0.03
Urban	4.00	0.65, 25.16	4.67	1.11, 19.64	
Population Density	1.00	0.99, 1.00	-	-	
Travel Time	1.00	1.00, 1.01	-	-	

*Results of the models are reported in odds ratios (OR) with corresponding 95% confidence intervals (CI).

doi:10.1371/journal.pntd.0002388.t002

compared to rural areas. Health seeking behaviors of urban and rural populations differ, which can bias reporting due to access to healthcare facilities. In the Caucasus region, rural populations were shown to under-utilize the national healthcare leading to an under reporting of illnesses [47]. Anthrax infections in livestock, as seen elsewhere may also prompt owners to slaughter animals and bring the meat to market more quickly [6]. Urban areas often function as the market centers, hence contaminated livestock brought to market may increase the risk of exposure. Also, occupational exposures may differ in and around urban areas with animal processing facilities close to central markets.

This study had several limitations that warrant discussion. The study was an ecological analysis that examined aggregate level data and may not represent the true associations of risk. Deriving census population as denominator data for small populations is often difficult, particularly in low resource settings. In this study, population totals for most communities were obtained from census records although several estimates were also obtained from gridded population data, which may skew the cumulative risk estimates and SaTScan results by either overestimating or underestimating the denominator figures. Model accuracy of the logistic regression was not evaluated with training and testing data due to the sample size of the persistent clusters identified. Although the disease was likely under reported, the reporting may have been biased towards urban areas with more access to healthcare facilities resulting in an under estimation of risk in rural areas. Furthermore, the method of classifying urban areas, while well documented in the literature, may in some instances incorrectly classify communities. Epidemiological data were not available on individuals, therefore individual level risk factors such as age, gender, and the source of infection were not considered in this study. Although anthrax is associated with specific factors on the landscape identified here, cultural and socio-economic factors not available in this study may also influence the occurrence of the disease.

Research has reiterated the need for continued surveillance in order to distinguish between an intentional release and natural occurring disease [56]. This is especially true in light of the 2001 anthrax bioterrorist attack in the United States, which has renewed fears of its use as a weaponized biological agent [18].

References

- Fasanella A, Galante D, Garofolo G, Hugh-Jones ME (2010) Anthrax undervalued zoonosis. Veterinary Microbiology 140: 318–331.
- Rasko DA, Worsham PL, Abshire TG, Stanley ST, Bannan JD, et al. (2011) Bacillus anthracis comparative genome analysis in support of the Amerithrax investigation. Proceedings of the National Academy of Sciences 108: 5027.
- Maudlin I, Eisler MC, Welburn SC (2009) Neglected and endemic zoonoses. Philosophical Transactions of the Royal Society B: Biological Sciences 364: 2777–2787.
- Swartz MN (2001) Recognition and management of anthrax—an update. New England Journal of Medicine 345: 1621–1626.
- Hugh-Jones M (1999) 1996–97 global anthrax report. Journal of applied microbiology 87: 189–191.
- WHO (2008) Anthrax in humans and animals. http://www.who.int/csr/ resources/publications/anthrax_web.pdf. 9241547537 9241547537.
- Hugh-Jones M, Blackburn J (2009) The ecology of *Bacillus anthracis*. Molecular Aspects of Medicine 30: 356–367.
- Blackburn JK, McNyset KM, Curtis A, Hugh-Jones ME (2007) Modeling the geographic distribution of Bacillus anthracis, the causative agent of anthrax disease, for the contiguous United States using predictive ecologic niche modeling. The American journal of tropical medicine and hygiene 77: 1103– 1110.
- Joyner TA, Lukhnova L, Pazilov Y, Temiralyeva G, Hugh-Jones ME, et al. (2010) Modeling the potential distribution of Bacillus anthracis under multiple climate change scenarios for Kazakhstan. PloS one 5: e9596.
- Mullins J, Lukhnova L, Aikimbayev A, Pazilov Y, Van Ert M, et al. (2011) Ecological Niche Modelling of the Bacillus anthracis A1. a sub-lineage in Kazakhstan. BMC ecology 11: 32.

Public health officials may be able to implement directed intervention strategies targeting high risk areas identified in this study which can include: training on proper carcass disposal, targeted livestock vaccination programs, education about the disease, and limiting occupational exposures through personal protective equipment [26,57,58]. While changing behavior related to zoonotic disease transmission may be difficult [26,59], initiating targeted livestock vaccination in areas with persistence risk can help to control future outbreaks. Future research should focus on collecting epidemiologic data on individual cases while also further exploring factors associated with an increasing rate of disease in communities identified in this study.

Supporting Information

Figure S1 Average incidence per 10,000 population of human cutaneous anthrax at the community within Georgia during the period 2000 to 2009. Cattle density is shown in two categories with areas in green representing \geq 19 head of cattle per square kilometer and lighter colored areas representing. (PDF)

′ re S2 Box plots sh

Figure S2 Box plots showing crude and Empirical Bayes smoothed (EBS) estimates for cumulative incidence for the period 2000 to 2004 and 2005 to 2009. (PDF)

Figure S3 Smoothed cumulative incidence for each 5 year period (top and middle) and the percent difference between the two periods (bottom) at the rayon level. (PDF)

Text S1 Methodology for district level mapping. (PDF)

Author Contributions

Conceived and designed the experiments: ITK JKB. Performed the experiments: ITK JKB. Analyzed the data: ITK LM. Contributed reagents/materials/analysis tools: LM NT JM LB PI ST. Wrote the paper: ITK JKB LM. Contributed and reviewed the manuscript and approved the final draft: ITK LM NT JM LB PI ST JKB.

- Peck RN, Fitzgerald DW (2007) Cutaneous anthrax in the Artibonite Valley of Haiti: 1992–2002. The American journal of tropical medicine and hygiene 77: 806–811.
- Cascio A, Bosilkovski M, Rodriguez-Morales A, Pappas G (2011) The socioecology of zoonotic infections. Clinical microbiology and infection 17: 336– 342.
- Pile JC, Malone JD, Eitzen EM, Friedlander AM (1998) Anthrax as a potential biological warfare agent. Archives of internal medicine 158: 429–437.
- Merabishvili M, Natidze M, Rigvava S, Brusetti L, Raddadi N, et al. (2006) Diversity of Bacillus anthracis strains in Georgia and of vaccine strains from the former Soviet Union. Applied and environmental microbiology 72: 5631– 5636.
- Imnadze P, Bakanidze L, Manrikyan M, Kukhalashvili T, Zangaladze E, et al. (2002) Anthrax in southern Caucasus. Antibiot Monit 18: 17–18.
- GIDEON. Anthrax in Georgia. Available: http://www.gideononline.com. Accessed 3/20/2013.
- Doganay M, Metan G (2009) Human anthrax in Turkey from 1990 to 2007. Vector-Borne and Zoonotic Diseases 9: 131–140.
- Chakraborty A, Khan SU, Hasnat MA, Parveen S, Islam MS, et al. (2012) Anthrax Outbreaks in Bangladesh, 2009–2010. The American journal of tropical medicine and hygiene 86: 703–710.
- Schuch R, Fischetti VA (2009) The secret life of the anthrax agent Bacillus anthracis: bacteriophage-mediated ecological adaptations. PLoS ONE 4: e6532.
- Kracalik IT, Blackburn JK, Lukhnova L, Pazilov Y, Hugh-Jones ME, et al. (2012) Analysing the spatial patterns of livestock anthrax in Kazakhstan in relation to environmental factors: a comparison of local (Gi*) and morphology cluster statistics. Geospatial Health 7: 111–126.

- Bakanidze L, Imnadz P, Tsanava S, Tsertsvadze N (2008) Disease Surveillance in Georgia: Benefits of International Cooperation. National Institute of Allergy and Infectious Diseases, NIH: 253–255.
- (NGA) (2012) NGIA GEOnet names server. Available: http://earth-infongamil/ gns/html/ Accessed 06/06/2012.
- 23. Index Mundi. Available: http://www.indexmundi.com/ Accessed 06/06/2012.
- 24. GeoNames Available: http://www.geonames.org/ Accessed 06/06/2012.
- Population Statistics of Éastern Europe Database. Available: http://pop-stat. mashke.org/ Accessed 06/06/2012.
- Jernigan JA, Stephens DS, Ashford DA, Omenaca C, Topiel MS, et al. (2001) Bioterrorism-related inhalational anthrax: the first 10 cases reported in the United States. Emerging infectious diseases 7: 933.
- GeoStat National Statistics Office of Georgia. Available: http://geostatge/ Accessed 06/20/2012.
- ESRI A (2009) 9.3. 1. Environmental Systems Research Institute, Redlands, United States.
- Anselin L, Syabri I, Kho Y (2006) GeoDa: An introduction to spatial data analysis. Geographical Analysis 38: 5–22.
- Waller LA, Gotway CA (2004) Applied spatial statistics for public health data: Wiley-Interscience.
- Moran PAP (1950) Notes on continuous stochastic phenomena. Biometrika 37: 17–23.
- Kulldorff M (2010) Information Management Services, Inc: SaTScanTM User Guide. SaTScan.
- Kulldorff M (1997) A spatial scan statistic. Communications in statistics-theory and methods 26: 1481–1496.
- Hijmans RJ, Cameron SE, Parra JL, Jones PG, Jarvis A (2005) Very high resolution interpolated climate surfaces for global land areas. International journal of climatology 25: 1965–1978.
- Hay SI, Tatem A, Graham A, Goetz S, Rogers D (2006) Global environmental data for mapping infectious disease distribution. Advances in parasitology 62: 37–77.
- Batjes N (2009) Harmonized soil profile data for applications at global and continental scales: updates to the WISE database. Soil Use and Management 25: 124–127.
- Nelson A (2008) Travel time to major cities: A global map of Accessibility. Global Environment Monitoring Unit-Joint Research Centre of the European Commission, Ispra Italy.
- Balk D, Deichmann U, Yetman G, Pozzi F, Hay S, et al. (2006) Determining global population distribution: methods, applications and data. Advances in parasitology 62: 119–156.
- Thornton P (2002) Mapping poverty and livestock in the developing world: ILRI (aka ILCA and ILRAD).
- Żweig MH, Campbell G (1993) Receiver-operating characteristic (ROC) plots: a fundamental evaluation tool in clinical medicine. Clinical chemistry 39: 561– 577.
- Bales ME, Dannenberg AL, Brachman PS, Kaufmann AF, Klatsky PC, et al. (2002) Epidemiologic Responses to Anthrax Outbreaks: A Review of Field Investigations, 1950–2001. Emerging Infectious Diseases 8: 1163.

- Gombe N, Nkomo B, Chadambuka A, Shambira G, Tshimanga M (2010) Risk factors for contracting anthrax in Kuwirirana ward, Gokwe North, Zimbabwe. African health sciences 10: 159–164.
- 43. Kartvelishvili T (2009) Cattle sector and dairy chain developments in Georgia, Azerbaijan and Armenia. The Cattle Sector in Central and Eastern Europe: Developments and Opportunities in a Time of Transition: 133.
- Macours K, Swinnen JFM (2008) Rural–urban poverty differences in transition countries. World Development 36: 2170–2187.
- Jones KE, Patel NG, Levy MA, Storeygard A, Balk D, et al. (2008) Global trends in emerging infectious diseases. Nature 451: 990–993.
- Sirisanthana T, Brown AE, Sirisanthana T, Brown A (2002) Anthrax of the gastrointestinal tract. Emerging infectious diseases 8: 649.
- Banda F, Chikampa W, Mutoloki S, Syakalima M, Munyeme M (2012) Risk analysis of an anthrax outbreak in cattle and humans of Sesheke district of Western Zambia. Acta Tropica 124: 162–5.
- Woods CW, Ospanov K, Myrzabekov A, Favorov M, Plikaytis B, et al. (2004) Risk factors for human anthrax among contacts of anthrax-infected livestock in Kazakhstan. The American journal of tropical medicine and hygiene 71: 48–52.
- Guh A, Heyman ML, Barden D, Fontana J, Hadler JL (2010) Lessons learned from the investigation of a cluster of cutaneous anthrax cases in Connecticut. Journal of Public Health Management and Practice 16: 201.
- Hampson K, Lembo T, Bessell P, Auty H, Packer C, et al. (2011) Predictability of anthrax infection in the Serengeti, Tanzania. Journal of Applied Ecology 48: 1333–1344.
- Aikembayev AM, Lukhnova L, Temiraliyeva G, Meka-Mechenko T, Pazylov Y, et al. (2010) Historical distribution and molecular diversity of Bacillus anthracis, Kazakhstan. Emerging Infectious Diseases 16: 789–796.
- Alexander KA, McNutt JW (2010) Human behavior influences infectious disease emergence at the human-animal interface. Frontiers in Ecology and the Environment 8: 522–526.
- Alexander KA, Lewis B, Marathe M, Eubank S, Blackburn JK (2012) Modeling of Wildlife Associated Zoonoses: Applications and Caveats. Vector-Borne and Zoonotic Diseases 12: 1005–18.
- Van Ness G, Stein C (1956) Soils of the United States favorable for anthrax. Journal of the American Veterinary Medical Association 128: 7.
- 55. Van Ness GB (1971) Ecology of anthrax. Science 172: 1303-1307.
- Clark DV, Ismayilov A, Bakhishova S, Hajiyev H, Nuriyev T, et al. (2011) Under-utilization of health care services for infectious diseases syndromes in rural Azerbaijan: A cross-sectional study. BMC health services research 11: 32.
- 57. Grunow R, Finke EJ (2002) A procedure for differentiating between the intentional release of biological warfare agents and natural outbreaks of disease: its use in analyzing the tularemia outbreak in Kosovo in 1999 and 2000. Clinical microbiology and infection 8: 510–521.
- Rotz LD, Hughes JM (2004) Advances in detecting and responding to threats from bioterrorism and emerging infectious disease. Nature medicine 10: S130– S136.
- Wein LM, Craft DL, Kaplan EH (2003) Emergency response to an anthrax attack. Proceedings of the National Academy of Sciences 100: 4346.